

# G OPEN ACCESS

**Citation:** Fornace KM, Senyonjo L, Martin DL, Gwyn S, Schmidt E, Agyemang D, et al. (2022) Characterising spatial patterns of neglected tropical disease transmission using integrated serosurveillance in Northern Ghana. PLoS Negl Trop Dis 16(3): e0010227. https://doi.org/10.1371/ journal.pntd.0010227

**Editor:** Eric HY Lau, The University of Hong Kong, CHINA

Received: February 1, 2021

Accepted: February 3, 2022

Published: March 8, 2022

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> <u>Commons CC0</u> public domain dedication.

Data Availability Statement: Relevant data are within this paper and its Supporting Information file or available through Senyonjo et. al, 2018, PLoS Neglected Tropical Diseases (https://doi.org/10. 1371/journal.pntd.0007027). Raw data are available online at: https://github.com/kfornace/ Ghana\_sero As spatial data on household locations are considered identifiable information, these data is only available for researchers who meet the criteria for access to confidential data and can be arranged by contacting Ethics@lshtm.ac.uk. RESEARCH ARTICLE

# Characterising spatial patterns of neglected tropical disease transmission using integrated sero-surveillance in Northern Ghana

Kimberly M. Fornace<sup>1,2\*</sup>, Laura Senyonjo<sup>3</sup>, Diana L. Martin<sup>4</sup>, Sarah Gwyn<sup>4</sup>, Elena Schmidt<sup>3</sup>, David Agyemang<sup>5</sup>, Benjamin Marfo<sup>6</sup>, James Addy<sup>6</sup>, Ernest Mensah<sup>7</sup>, Anthony W. Solomon<sup>1,8</sup>, Robin Bailey<sup>1</sup>, Chris J. Drakeley<sup>1</sup>, Rachel L. Pullan<sup>1</sup>

1 Faculty of Infectious and Tropical Diseases, London School of Hygiene & Tropical Medicine, London, United Kingdom, 2 Institute of Biodiversity, Animal Health and Comparative Medicine, University of Glasgow, Glasgow, United Kingdom, 3 Research Team, Sightsavers UK, Haywards Heath, United Kingdom, 4 Division of Parasitic Diseases and Malaria, Centers for Disease Control and Prevention, Atlanta, Georgia, United States of America, 5 Ghana Office, Sightsavers, Accra, Ghana, 6 Neglected Tropical Disease Team, Ghana Health Service, Accra, Ghana, 7 FHI 360, Accra, Ghana, 8 Department of Control of Neglected Tropical Diseases, World Health Organization, Geneva, Switzerland

\* Kimberly.Fornace@lshtm.ac.uk

# Abstract

## Background

As prevalence decreases in pre-elimination settings, identifying the spatial distribution of remaining infections to target control measures becomes increasingly challenging. By measuring multiple antibody responses indicative of past exposure to different pathogens, integrated serological surveys enable simultaneous characterisation of residual transmission of multiple pathogens.

## Methodology/Principal findings

Here, we combine integrated serological surveys with geostatistical modelling and remote sensing-derived environmental data to estimate the spatial distribution of exposure to multiple diseases in children in Northern Ghana. The study utilised the trachoma surveillance survey platform (cross-sectional two-stage cluster-sampled surveys) to collect information on additional identified diseases at different stages of elimination with minimal additional cost. Geostatistical modelling of serological data allowed identification of areas with high probabilities of recent exposure to diseases of interest, including areas previously unknown to control programmes. We additionally demonstrate how serological surveys can be used to identify areas with exposure to multiple diseases and to prioritise areas with high uncertainty for future surveys. Modelled estimates of cluster-level prevalence were strongly correlated with more operationally feasible metrics of antibody responses.

### **Conclusions/Significance**

This study demonstrates the potential of integrated serological surveillance to characterise spatial distributions of exposure to multiple pathogens in low transmission and elimination settings when the probability of detecting infections is low.

Funding: Sample collection was funded by The Coalition for Operational Research of Neglected Tropical Diseases (COR-NTD), funded through The Task Force for Global Health by the Bill and Melinda Gates Foundation, United Kingdom Department for International Development and the United States Agency for International Development through its Neglected Tropical Diseases Program (LS). Additional funding for sample analysis was provided by the Bill and Melinda Gates Foundation (RLP), Sightsavers (LS) and the Centers for Disease Control and Prevention (DLM, SG). KMF is supported by a Sir Henry Dale Fellowship jointly funded by the Wellcome Trust and the Royal Society (Grant Number 221963/Z/20/Z). AWS is a staff member of the World Health Organization. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

**Disclaimer:** The authors alone are responsible for the views expressed in this article and they do not necessarily represent the views, decisions or policies of the institutions with which they are affiliated.

#### Author summary

Following implementation of successful interventions, one of the primary challenges for neglected tropical disease programmes is identifying areas with remaining disease transmission. As disease prevalence decreases, these infections become increasingly rare and hard to detect. Serological assays measure long-lived disease-specific antibody responses indicating past exposure to pathogens and increase the probability of detecting disease transmission. Here, we integrate serological assays with environmental and spatial data to map priority areas for surveillance for multiple neglected tropical diseases in Northern Ghana using the two-stage cluster-based survey platform established for trachoma surveillance. The use of multiplex bead assays measuring exposure to multiple pathogens allows integrated surveillance of diseases of interest to the national control programme. We identify areas with high risks of transmission to selected diseases as well as areas with high uncertainty which are priorities for future control and surveillance efforts. Together, this highlights the utility of multiplex serological platforms as a tool for integrated surveillance and mapping of neglected tropical diseases.

#### Introduction

Neglected tropical diseases (NTDs), such as trachoma, schistosomiasis, onchocerciasis, lymphatic filariasis and soil-transmitted helminthiases, cause substantial public health burdens globally. With increasing investment in NTD control, elimination and eradication programmes, community-based surveys are used to monitor impacts of interventions, identify residual transmission and target high-risk populations [1,2]. These data are frequently analysed using geostatistical models, relating infection metrics to environmental and spatial covariates to define the geographical extent of transmission, predict disease burdens and prioritise areas of uncertainty [3,4]. However, as countries move towards elimination, infections become increasingly rare and difficult to monitor through community-based surveys. Such challenges are currently faced by NTD control programmes operating in Northern Ghana where trachoma elimination was achieved in 2018 and only isolated transmission is reported for onchocerciasis and other NTDs [5,6]. New approaches are needed to cost-effectively identify remaining infections and areas at high risk of recrudescence.

As transmission of infectious diseases decrease, the probability of detecting infections becomes correspondingly low and requires prohibitively large sample sizes to identify infections. For many diseases, this also corresponds with more pronounced spatial heterogeneity, with transmission concentrated in specific geographic areas or sub-populations [7–9]. Serological assays are potentially sensitive tools in these contexts; by measuring specific antibody responses reflecting previous exposure to pathogens, changes in transmission can be detected over longer durations in higher proportions of populations [10]. This enables estimation of the force of infection and historical transmission intensities using age-stratified antibody magnitudes or longitudinal sampling, allowing exploration of control measure impacts over time [11–14]. These sero-epidemiological methods have been applied to NTDs including trachoma, lymphatic filariasis, schistosomiasis, and onchocerciasis as well as enteropathogens and are strongly correlated with commonly used metrics of infection prevalence (e.g. [13,15–17]).

The development of multiplex serological assays increases the operational feasibility of serological techniques by enabling measurement of a broad range of responses with high repeatability from limited blood samples. Multiplex bead assays (MBA) are a well-validated method to measure population-level serological responses to a wide panel of NTDs, malaria, vaccine preventable diseases and other infections [18]. These assays allow measurement of antibody responses to multiple pathogens simultaneously for relatively low cost, creating new opportunities for integrated serological surveillance and maximising the benefit of limited resources [19,20]. Applying these integrated approaches can detect populations at high risks of multiple infections, identifying public health gaps and allowing targeting of coordinated interventions. Further, analysis of these data within geostatistical modelling frameworks provides new opportunities to characterise the spatial distribution of multiple diseases and identify environmental factors driving residual transmission [21]. Geostatistical modelling typically also allows greater precision of estimates using fewer data points than conventional analytical approaches [22].

Despite increasing collection of serological data, challenges remain in translating these data into actionable programmatic information [23]. Spatial analysis of serological data is complicated by the duration of antibody responses and frequency of exposure to infection; while it is possible to map seropositivity, resulting maps may reflect historical exposure rather than current transmission and have limited utility to programmes. Additionally, applying commonly used geostatistical modelling approaches to quantitative antibody responses may not capture outlying values of high responders, likely representing individuals recently or repeatedly exposed to a pathogen. A recent study demonstrated how the force of infection calculated from paired serological surveys could be used to quantify spatial heterogeneity of schistosomiasis transmission, showing close correlation with other routinely collected measures of schistosomiasis infection intensity [24]. While highlighting the utility of serological assays to characterise spatial patterns of transmission, this modelling approach utilises surveys of the same area over multiple time points to quantify transmission over time, data not routinely collected by NTD programmes. Alternatively, for diseases such as malaria and cholera, detailed longitudinal cohort studies have identified markers of recent exposure or been used to train machine learning approaches to estimate sero-incidence; however, these cohort data are not available for most NTDs [25,26].

Here, we adapt a commonly used Gaussian mixture model and binomial geostatistical models to describe the spatial distribution of antibody responses to multiple diseases in a population-based survey of children in Northern Ghana. Conducted as part of routine surveillance for trachoma elimination, this survey utilised a two-stage cluster-sampling population-based survey design and identified very low levels of the sign "trachomatous inflammation—follicular", a finding supported by estimates of seroconversion rates [27–29]. Although this low prevalence precludes modelling the spatial distribution of infection, geostatistical modelling of serological data allows prioritisation of geographical areas of potential risk of recrudescence that can be targeted for future trachoma post-validation surveillance efforts. As this survey used a systematically- and randomly- sampled population-based approach rather than purposeful sampling, assessment of spatial patterns of other disease transmission is possible. Using multiplex serological data, we demonstrate how an integrated surveillance approach can be used to estimate the geographical distribution of exposure to other pathogens within this population and identify areas with previously unknown elevated risks of multiple diseases requiring prioritisation for national surveillance programmes.

#### Methods

#### **Ethics statement**

This study was approved by the Ghana Health Service Ethics Review Committee (GHS-ERC: 03/07/15) and the London School of Hygiene & Tropical Medicine (10285). Written informed consent was sought from parents or guardians of all participating children. Verbal assent was

additionally obtained from children who were able to provide this. The CDC investigators were not considered to be engaged in human subjects research.

#### Survey and laboratory methods

This study was conducted in the Northern, North East, Savanna and Upper West regions of Ghana in a predominantly rural agricultural population (Fig 1). The climate is tropical and elevations range from sea level to 900 metres above sea level. This area includes several national parks and multiple rivers, including tributaries to the Volta [27]. From 2015–2016, two-stage cluster-sampled population-based surveys were conducted as part of trachoma elimination pre-validation activities [27]. Briefly, clusters (villages) were sampled with probability proportional to size and households selected using compact segment sampling across 18 evaluation units. All children aged 1–9 years were eligible for inclusion. These ages were targeted as younger children have the highest risk of active trachoma and the primary purpose of the survey was to provide evidence for the validation of elimination of trachoma in Ghana. Basic demographic data and GPS coordinates of household locations were collected electronically using Open Data Kit (www.getodk.org).

All children from consenting households provided a finger-prick blood sample, which was collected and stored on filter paper (Trop-Bio, Townsville, Australia) and processed at the U.S. Centers for Disease Control and Prevention as previously described [28]. Using a multiplex bead assay (MBA), immunoglobulin G antibody responses were measured to the following antigens: Pgp3 trachoma (*Chlamydia trachomatis*), Wb123 lymphatic filariasis (*Wucheria bancrofti*), Ov16 onchocerciasis (*Onchocerca volvulus*), NIE strongyloidiasis (*Strongyloides* 



Fig 1. Study site in Northern Ghana; administrative shapefiles obtained from National Information Technology Agency (NITA), Government of Ghana (https://data.gov.gh/dataset/shapefiles-all-districts-ghana-170-districts). The boundaries and names shown and the designations used on this map do not imply the expression of any opinion whatsoever on the part of the authors, or the institutions with which they are affiliated, concerning the legal status of any country, territory, city or area or of its authorities, or concerning the delimitation of its frontiers or boundaries.

https://doi.org/10.1371/journal.pntd.0010227.g001

*stercoralis*), soluble egg antigen (SEA) schistosomiasis (*Schistosoma mansoni*), rp17 and TmpA yaws (*Treponema pallidum*), VSP3 giardiasis (*Giardia lamblia*). Glutathione s-transferase (GST) was used as a negative control. MBA were conducted using standard methods [28]. Antibody responses were quantified as median fluorescence intensity (MFI) for each antigen and sample. We excluded samples that had an MFI value for GST over 1000 (3 samples excluded, median GST MFI = 4, IQR: 2–7), which indicates high background levels and unreliable measurements.

#### Classification of seropositivity

In the absence of standard controls or clinically characterised sera to determine cut-off values, Gaussian mixture models are commonly used to determine seropositivity, modelling antibody responses as latent populations for a user-defined number of distributions. These distributions can then be used to define antigen-specific cut-off values dependent on the number of observations and ranges of values. While these algorithms are frequently applied to differentiate seronegative and seropositive populations, this approach can be extended to include more than two components, with components corresponding to different degrees of pathogen exposure [30]. As programmes require data on the distribution of recent exposure operationally, and environmental and spatial data reflected current conditions, we fit a three-component Gaussian mixture model to log-transformed antibody data. We assumed these components corresponded to unexposed, historically exposed and recently or repeatedly exposed populations [31]. While antibody responses may have differing numbers of components, we chose to use a three-component mixture model specifically to identify individuals with the highest MFI responses likely to be recently exposed. Other approaches, such as modelling continuous antibody response data, may characterise mean antibody response levels but are unlikely to accurately capture outliers. Similarly, approaches using a predefined centile of the highest responders may identify the individuals with highest likelihoods of recent exposure but cannot be used to assess prevalence as the percentage positive is predetermined [32]. Using an approach described for enteropathogens, where high intensity and repeated exposure is observed in young children, we fit mixture models to children aged under 3 years to ensure sufficient proportions of unexposed individuals that would enable characterisation of the seronegative population [15]. Mixture models were informed by priors on the expected proportions of individuals in each component based on published data and consultations with the NTD programme (Table A in <u>S1 Text</u>). For example, we assumed a low prior probability of recent or historical yaws due to previous reports of elimination in this area [33]. For geostatistical modelling, we defined seropositive individuals as individuals belonging the highest component and most likely to have recent or repeated exposure.

#### Environmental and spatial data

We assembled plausible spatial and environmental covariates including population density [34], travel time to the nearest city [35], insecticide treated bednet coverage, soil types [36], forest cover [37] and bioclimatic factors representing long-term and more recent historical annual trends, seasonality and extreme or limiting environmental factors [38]. Topographic wetness index was calculated from a digital elevation model [39] and Euclidean distance to roads and water bodies were calculated at 30m resolution from Open Street Map data [40] (Table A in S1 Text). Normalised difference vegetation index and normalised difference water index were calculated from cloud-free composites of Landsat data during the survey time period [41]. Pearson correlation analysis was used to assess multicollinearity between variables, excluding variables with correlation coefficients > 0.75. Models were fit separately to assess exposure to each pathogen and covariates were selected based on the Bayesian information criteria (BIC) of the most parsimonious non-spatial binomial regression model for each disease and residual spatial autocorrelation was assessed using Moran's I. To generate covariates for predictions, we resampled all data to 1000m resolution and extracted values. We excluded all national parks and protected areas [42].

#### Geostatistical modelling

Final models were developed within a Bayesian hierarchical modelling framework. In addition to including identified covariates, these models also used a geostatistical modelling approach to account for residual spatial autocorrelation not explained by these covariates. For each disease, we fit separate models with  $p(x_i)$  denoting the seroprevalence at locations  $x_i$ , i = 1...n, with  $m_i$  individuals sampled per location. The full model was specified as:

$$Y_i \sim \text{Binomial}(m_i, p(x_i))$$

With the linear predictor for the binomial model specified as:

$$logit(p(x_i)) = \beta_0 + d(x_i)'\beta + w$$

Where  $\beta_0$  represents the intercept,  $d(x_i)'\beta$  represents a vector of location specific covariate effects and  $w_i$  represents the spatial effect. Weakly informative priors of Normal (0, 100) were used for intercepts and fixed effect coefficients and penalised complexity priors were used for the spatial effect [43]. As Moran's I showed significant spatial autocorrelation for all diseases modelled, we additionally fit models with the spatial effect modelled as a Matern covariance function between locations  $s_i$  and  $s_k$ :

 $W \sim Multivariate Normal (0, \Sigma)$ 

$$\Sigma_{jk} = Cov\left(\xi\left(s_{j}\right), \xi(s_{k})\right) = Cov\left(\xi_{j}, \xi_{k}\right) = \frac{\sigma^{2}}{\Gamma(\lambda)2^{\lambda-1}} \left(\kappa||s_{i} - s_{k}||\right)^{\lambda} K_{\lambda}\left(\kappa||s_{j} - s_{k}||\right)$$

Final models were assessed using the deviance information criteria (DIC) and conditional predictive ordinate (CPO). All models were implemented in R statistical software version 3.6 using Integrated Nested Laplace Approximation (INLA), using 1,000 samples to estimate posterior probabilities [44].

To prioritise areas for future sampling, we calculated exceedance probabilities for a threshold of 10% seroprevalence of high responders. We chose a threshold of 10% as this indicates on-going transmission in most settings, although this threshold could be adjusted based on programmatic requirements. These exceedance probabilities represent the probability that an area has over 10% seroprevalence, enabling visualisation of high-risk areas as well as identification of areas with high uncertainty [45]. Priority areas with high probabilities of multiple diseases were identified by overlaying classified exceedance probabilities. To define areas with high probabilities of recent exposure, we classified high-risk areas as areas with 70% or greater probabilities of exceeding the 10% seroprevalence threshold. We additionally classified areas where further data are needed to estimate serological exposure. These areas were classified as having exceedance probabilities between 40–60%, indicating further data are needed to determine whether these areas are above or below this threshold.

As fitting geostatistical models is likely not feasible for most control programmes, we compared estimates of seroprevalence derived from geostatistical models to more simple metrics of exposure which included the arithmetic mean, geometric mean, median, coefficient of variation and standard deviation of cluster MFI values. To evaluate these, we calculated the mean posterior estimates of seroprevalence for all households within each cluster from geostatistical models. Exploratory analysis identified the strongest correlation between arithmetic mean and seroprevalence. To further assess this metric, relationships between estimated seroprevalence and arithmetic mean MFI values per cluster were explored using B-spline regression, with optimal degrees selected using BIC.

#### Results

A total of 10993 children from 201 clusters across 9 districts were sampled between November 2015 and April 2016. Data on 154 children were excluded due to high background or incomplete GPS data, resulting in a final dataset of 10840 children from 3444 households. The median age of children included was 5 years (IQR: 3–7), with roughly equal numbers of male and female children included (50.3% male, n = 5531). Distributions of antibody responses varied by pathogen (Fig A in S1 Text). Age-specific patterns of antibody responses were markedly different between antigens (Fig A in S1 Text). Antibody responses for VSP3, SEA and NIE all showed high responses in very young age groups, consistent with previous literature describing high exposure in very young children [15]. To further explore age-specific antibody responses, we compared densities between different age groups (Fig B in S1 Text). Using mixture models, we estimated seroprevalence and identified the distribution of MFI responses within the highest component (Table 1).

Using these data, we identified environmental and spatial risk factors for household seroprevalence of high responders for each disease. As the seroprevalence for yaws was very low, consistent with available data from this region [33], we excluded these antigens from spatial modelling. Final models identified different effects of covariates, with most diseases showing some association with climatic factors (Table 2). Varying spatial effects were observed across diseases; however, model fit was improved by the inclusion of spatial terms for all diseases.

Within Northern Ghana, we predicted the probability of exceeding the 10% seropositivity at 1000m resolution (Fig 2). Using these exceedance probabilities, we identified regions with high probabilities of recent transmission of multiple diseases (Fig 3A). We also found areas with high probabilities of multiple diseases, such as at the western border of the study area; this may indicate poor coverage by health programmes, human movement from neighbouring countries or other risk factors. We additionally identified priority areas for future surveys based on high uncertainty; these were defined as having exceedance probabilities between 40 and 60%, indicating models were not able to determine whether these regions were above or below 10% seropositivity (Fig 3B).

Table 1.	Estimates of seroprevalence b	ov antigen and disease,	including distributions	s of responses for all ind	ividuals and individuals	classified as exposed.
	1	1 0 1		1		1

			MFI (median, IQR)		
Disease	Antigen	Exposed (p, %)	All individuals	Exposed	
Trachoma	Pgp3	201 (1.9%)	10 (6-20)	9870 (2517–26115)	
Lymphatic Filariasis	Wb123	54 (0.5%)	161 (79–326)	5688 (3633–9096)	
Onchocerciasis	Ov16	116 (1.1%)	5 (3–7)	1002 (486–4294)	
Strongyloidiasis	NIE	278 (2.6%)	43 (24-81)	865 (530–2192)	
Schistosomiasis	SEA	1460 (13.5%)	275 (143-531)	1029 (871–1302)	
Giardiasis	VSP3	967 (8.9%)	315 (115-1219)	15630 (10424–23933)	
Yaws	rP17	19 (0.2%)	16 (11–25)	1842 (1208–20565)	
	TmpA	8 (0.1%)	6 (4-9)	3722 (467–13088)	

https://doi.org/10.1371/journal.pntd.0010227.t001

Covariate*	95% BCI		
	Mean	2.5%	97.5%
a. Trachoma Pgp3			
Annual precipitation (mm)	-0.568	-1.079	-0.066
Mean precipitation of warmest quarter (1970–2000), (mm)	0.182	-0.100	0.454
Mean precipitation of coldest quarter (1970–2000), (mm)	0.999	0.274	1.735
Mean temperature (1970–2000), (°C)	-1.225	-1.754	-0.709
Minimum temperature (1970–2000), (°C)	1.155	0.450	1.861
Accessibility, (hours to city)	0.106	-0.117	0.319
Soil type (water pH at 0cm)	0.252	-0.002	0.507
Spatial range (km)	17.416	8.016	34.97
b. Lymphatic filariasis Wb123			
Mean temperature (1970–2000), (°C)	-1.083	-1.692	-0.403
Minimum temperature (1970–2000), (°C)	1.008	0.303	1.601
Land surface temperature at night, (2002–2017), (°C)	-0.338	-0.739	0.036
Topographic wetness index	0.289	-0.085	0.637
Population density	-0.189	-0.696	0.254
Distance to roads (km)	0.176	-0.096	0.413
Soil type (water pH at 0cm)	0.776	0.291	1 255
Spatial range (km)	43.01	9.852	1.255
Spatial range (Kill)	45.01	0.032	141.02
c. Onchocerclasis OV16	0.249	0.922	0.147
Annual precipitation (mm)	-0.348	-0.833	0.14/
Land surface temperature at hight, (2002–2017), (C)	-0.686	-1.046	-0.344
1 opographic wetness index	-0.343	-0./21	0.013
Accessibility, (hours to city)	0.501	0.010	0.978
Soil type (proportion of silt at 0cm)	0.310	-0.040	0.666
Spatial range (km)	29.83	14.67	55.17
d. Schistosomiasis SEA			
Mean precipitation of warmest quarter (1970–2000), (mm)	0.528	0.347	0.710
Mean precipitation of coldest quarter (1970–2000), (mm)	0.619	0.439	0.800
Mean temperature, (1970–2000), (°C)	0.620	0.440	0.799
Distance to water bodies, (km)	-0.167	-0.347	0.012
Normalised difference vegetation index	0.214	0.104	0.325
Distance to roads, (Km)	-0.381	-0.564	-0.197
Spatial range (km)	7.102	6.420	8.054
e. Strongyloidiasis NIE			
Annual precipitation (mm)	0.326	0.037	0.615
Mean precipitation of warmest quarter (1970–2000), (mm)	-0.245	-0.525	0.035
Mean precipitation of coldest quarter (1970–2000), (mm)	-0.575	-0.890	-0.260
Minimum temperature, (1970–2000), (°C)	-0.222	-0.506	0.061
Insecticide treated net coverage	0.213	-0.062	0.488
Population density	-0.100	-0.209	0.009
Distance from reads (lim)	-0.158	-0.36/	0.050
	0.349	-0.059	0./5/
Spatial range (Km)	26.235	3.458	98.559
r. Giardia VSP3	0	0.000	0.677
Annual precipitation (mm)	0.113	-0.032	0.257
Mean precipitation of warmest quarter (1970–2000), (mm)	-0.180	-0.304	-0.056

Table 2. Mean posterior estimates of coefficients of fixed effects and spatial range, including 95% Bayesian credible intervals (BCI).

(Continued)

Table 2. (Continued)

Covariate*	95% BCI		
	Mean	2.5%	97.5%
Mean precipitation of coldest quarter (1970–2000), (mm)	-0.332	-0.498	-0.166
Land surface temperature at night, (2002–2017), (°C)	-0.107	-0.211	-0.003
Minimum temperature, (1970–2000), (°C) Normalised difference vegetation index	-0.137 -0.111	-0.288 -0.200	0.013 -0.021
Spatial range (km)	12.315	2.880	33.69

\* All covariates mean-centred and scaled

https://doi.org/10.1371/journal.pntd.0010227.t002

When simple metrics of cluster MFI values were compared to the exceedance probabilities for all households within the cluster, the arithmetic mean MFI showed the strongest correlation with mean posterior estimates of cluster seroprevalence. Heterogeneity between MFI values allowed identification of high-risk clusters (Fig 4). Relationships between these metrics were nonlinear and highly variable for all diseases surveyed, with arithmetic mean MFI values closely associated with cluster-level seroprevalence for high burden diseases (e.g. schistosomiasis) but less correlated for lower burden diseases (Fig C in S1 Text). However, for all diseases surveyed, prioritising clusters with the highest mean MFI values would enable targeting clusters with the highest probabilities of recent exposure.

#### Discussion

This study demonstrates the utility of geostatistical modelling of integrated serological surveys to characterise spatial patterns of NTDs and other pathogens. Despite the increasing use of this



**Fig 2.** Posterior estimates of the probability of exceeding 10% seroprevalence of high responders to A) Trachoma Pgp3; B) Lymphatic filariasis Wb123; C) Onchoceriasis Ov16; D) Strongyloidiasis NIE; E) Schistosomiasis SEA; F) Giardiasis VSP3; administrative shapefiles obtained from National Information Technology Agency (NITA), Government of Ghana (https://data.gov.gh/dataset/shapefiles-all-districts-ghana-170-districts).

https://doi.org/10.1371/journal.pntd.0010227.g002



**Fig 3.** Combining disease measures to A) identify regions with high probabilities of exceeding 10% seroprevalence (exceedance probabilities > 70%); and B) identify regions with high uncertainty (exceedance probabilities 40–60%); administrative shapefiles obtained from National Information Technology Agency (NITA), Government of Ghana (https://data.gov.gh/dataset/shapefiles-all-districts-ghana-170-districts).

https://doi.org/10.1371/journal.pntd.0010227.g003

MBA, to our knowledge, this is the first study integrating multi-disease data within geostatistical frameworks. By estimating proportions of individuals with elevated antibody levels suggesting recent exposure, we demonstrate how serological data can be combined with contemporaneous spatial and environmental data. These geostatistical models of serological data broadly reflect known disease status, while additionally highlighting some previously unknown foci of possible transmission. This could become a valuable tool for control programmes operating in low transmission settings where probabilities of detecting infections are extremely low. In these cases, analysis of serological data within geostatistical frameworks enables identification of environmental and spatial risk factors associated with residual transmission and allows prioritisation of areas for future control and surveillance efforts. Further, we demonstrate how simple metrics of antibody responses can provide actionable information to control and elimination programmes to prioritise areas with high probabilities of transmission.

Although a large body of literature explores the utility of serological assays as measures of transmission, these data are rarely analysed within geostatistical frameworks and then primarily for single diseases (e.g. [21,46]). This is in part due to the challenges linking serological data, which may represent past exposure, with spatial and environmental data representing current locations and conditions. While in some cases, simple maps of serological data may be





https://doi.org/10.1371/journal.pntd.0010227.g004

sufficient to delineate the boundaries of transmission or monitor infection introductions, identification of areas of on-going transmission of a previously endemic disease requires estimation of recent exposure. There remains no standard approach for classifying seropositivity from these MBAs, with reference values typically only available for vaccine preventable diseases [19]. Similar to previous studies, we identified very high rates of exposure to entero-pathogens in young children, necessitating fitting mixture models using subsets of the youngest children to ensure sufficient unexposed individuals [15]. In the absence of longitudinal data on antibody kinetics, identifying high responses can act as a proxy for recent or repeated exposure. Within this study, a three-component mixture model was used to identify the highest responders for each antigen. While this has the advantage of identifying individuals with the highest antibody titres and priorities for future surveillance, estimates of seropositivity should be interpreted cautiously as this may not represent all exposed individuals. Similarly, public health implications of groups identified with high responses may vary by disease and epidemiology within this context. If more detailed longitudinal measurements were available, future studies could explore mechanistic modelling of antibody dynamics to more accurately identify exposure groups (e.g. [47,48]). However, the correlation between serological estimates and other metrics of disease transmission strongly supports the use of these assumptions.

Within this setting, serological data support other estimates of NTD burdens and associated environmental risk factors within Northern Ghana. These results are consistent with previous findings of very low prevalence of trachoma and additionally demonstrate how serology may be employed to identify areas for post-elimination surveillance for potential recrudescence [27,28]. Additionally, analysis of serological data on exposure to endemic pathogens can be used to prioritise areas with high probabilities of recent transmission or high uncertainty for one or multiple diseases, to identify targets for future control or surveillance efforts. In particular, examination of areas with high uncertainty reveals key knowledge gaps. For example, despite strong associations with distance from water and other environmental predictors, household-level prevalence of schistosomiasis is unknown for most of the study sites due to the extremely high focality of the disease. These levels of focality may not be represented by school-based or other indicator surveys not capturing the full distribution of schistosomiasis within the community. The uncertainty in model results also reflects the distribution of sampling points; visualisation of this uncertainty allows prioritisation of areas for future surveys. This has important implications both when planning surveillance and when allocating treatment at larger administrative units, as is commonly done for mass drug administration [49].

Geostatistical modelling of serological data representing exposure to multiple pathogens has important implications for survey design. The systematic and random population-based sampling strategy used for trachoma mapping enabled assessment of the distribution of other NTDs; this approach would not have been feasible with purposive sampling. For example, this study identified a focus of onchocerciasis with a significant prevalence of anti-Ov16 antibodies in children aged 1–9 years in Saboba-Cherepon. Previous mapping of onchocerciasis in this area through the Rapid Epidemiological Mapping of Onchocerciasis (REMO) in 2009 determined this to be an area of hypo-endemicity, not requiring onchocerciasis specific MDA at that time. Since then, the area has been treated for lymphatic filariasis, which includes the delivery of ivermectin (the drug also given in MDA for onchocerciasis) and the expectation had been this would also impact on O. volvulus transmission. However, lymphatic filariasis MDA ceased in 2014, and these serological data indicate an increase in onchocerciasis previously unknown to GHS and as a direct result of the integrated surveillance approach, GHS decided to start MDA specifically for onchocerciasis in this focus (personal communication, Ghana Health Service). While further work is required to confirm transmission and disease burdens may have changed since 2015, this identifies a priority area for surveillance.

For control and elimination programmes needing to make operational decisions quickly without access to technical expertise needed to develop geostatistical models, very simple metrics can be applied to identify locations at high risk of transmission. While the relationship between arithmetic mean MFI values and estimated seroprevalence is not linear, prioritising control activities at clusters with the highest mean MFI values would likely ensure that interventions are reaching communities with the highest seroprevalence. This agrees with previous studies showing high correlation between mean quantitative antibody levels and other infection-based metrics of NTD transmission [14,24]. As serological samples can be collected using finger-prick blood sampling with no need for cold chains in the field, this is an operationally attractive and cost-effective method compared to many other diagnostics. This utility is further increased by the ability to multiplex, allowing information to be collected for multiple diseases of public health importance [19,50]. Ideally, future studies could compare metrics obtained by integrated serological surveys with routinely used diagnostics to enable more efficient application for disease mapping surveys and treatment decisions.

Despite the value of this approach, this study had several important limitations. As this study does not include adults, further validation of these methods may be required in settings with very high historical transmission and exposure in older age groups. While serological surveys of children are likely to be a better marker of recent rather than historical exposure, the utility of this approach will vary for specific diseases and associated risk factors. For trachoma, young children under 9 are believed to be the primary source of infection. In contrast, children aged under 5 years have lower risks of onchocerciasis and schistosomiasis, which predominantly impact school-aged children and adults with high-risk occupational activities. This highlights a key methodological challenge of integrated surveys targeting multiple diseases with different risk groups and transmission mechanisms. Additionally, this analysis relied on a single dataset collected over one time point. These diseases may additionally have differing immune responses, antibody kinetics and infection periods and longitudinal data may be able to more accurately identify recent infections or characterise transmission [48]. The uncertainty identified within these models may be further reduced by inclusion of other data and surveillance information and cannot be considered a reflection of the current knowledge of disease control across this region. However, the Bayesian framework used can be easily extended to incorporate other sources of information, such as pairing serological data with infection data [51].

Despite these limitations, this study demonstrates how integrated serological surveillance can characterise the spatial distribution of exposure to multiple pathogens. An adaptable framework is provided to understand the spatial and environmental factors driving transmission in elimination settings when infection data are rare. As countries approach elimination, this study additionally highlights the need for innovative surveillance approaches utilising population representative sampling to maximising efficiency by collecting data across multiple diseases. Applying these techniques can provide valuable information for control programmes needing to identify and target remaining foci of infection.

#### **Supporting information**

**S1 Text. Supplementary Information file. Table A.** Priors used for proportions in three component mixture models. **Table B.** Spatial and environmental covariates. **Fig A.** Disease-specific antibody responses for all included children, i. Density plots of log-transformed MFI values, ii. Age distributed antibody responses. **Fig B.** Antibody densities by age categories. **Fig C.** Relationships between mean posterior estimates of seroprevalence and arithmetic mean MFI per cluster for: A) Trachoma; B) Filariasis; C) Onchocerciasis; D) Strongyloides; E)

Schistosomiasis; F) Giardiasis. (DOCX)

#### Acknowledgments

We would like to thank the survey teams, study participants and Upper West and Northern Regional Health Directorates of Ghana for their contributions to this study. We would additionally like to acknowledge Dr. Oscar Debrah, the former Head of the Eye Care Unit of the Ghana Health Service for supporting data collection activities.

#### **Author Contributions**

Conceptualization: Laura Senyonjo, Anthony W. Solomon, Robin Bailey.

**Data curation:** Laura Senyonjo, David Agyemang, Benjamin Marfo, James Addy, Ernest Mensah.

Formal analysis: Kimberly M. Fornace, Laura Senyonjo.

- Funding acquisition: Laura Senyonjo, Rachel L. Pullan.
- **Investigation:** Kimberly M. Fornace, Laura Senyonjo, Diana L. Martin, Sarah Gwyn, Elena Schmidt, David Agyemang, Benjamin Marfo, James Addy, Ernest Mensah.
- Methodology: Kimberly M. Fornace, Diana L. Martin, Sarah Gwyn, Elena Schmidt, Benjamin Marfo, James Addy, Ernest Mensah, Chris J. Drakeley.
- Supervision: Benjamin Marfo, James Addy, Rachel L. Pullan.

Validation: Diana L. Martin, Chris J. Drakeley.

Visualization: Kimberly M. Fornace.

- Writing original draft: Kimberly M. Fornace, Laura Senyonjo.
- Writing review & editing: Kimberly M. Fornace, Laura Senyonjo, Diana L. Martin, Sarah Gwyn, Elena Schmidt, David Agyemang, Benjamin Marfo, James Addy, Ernest Mensah, Anthony W. Solomon, Robin Bailey, Chris J. Drakeley, Rachel L. Pullan.

#### References

- Engels D, Zhou XN. Neglected tropical diseases: an effective global response to local poverty-related disease priorities. Infect Dis Poverty. 2020; 9(1):10. https://doi.org/10.1186/s40249-020-0630-9 PMID: 31987053; PubMed Central PMCID: PMC6986060.
- Baker MC, Mathieu E, Fleming FM, Deming M, King JD, Garba A, et al. Mapping, monitoring, and surveillance of neglected tropical diseases: towards a policy framework. Lancet. 2010; 375(9710):231–8. https://doi.org/10.1016/S0140-6736(09)61458-6 PMID: 20109924.
- Stanton MC. The Role of Spatial Statistics in the Control and Elimination of Neglected Tropical Diseases in Sub-Saharan Africa: A Focus on Human African Trypanosomiasis, Schistosomiasis and Lymphatic Filariasis. Advances in parasitology. 2017; 97:187–241. https://doi.org/10.1016/bs.apar.2017.01. 001 PMID: 28325371.
- Pigott DM, Howes RE, Wiebe A, Battle KE, Golding N, Gething PW, et al. Prioritising Infectious Disease Mapping. PLoS neglected tropical diseases. 2015; 9(6):e0003756. https://doi.org/10.1371/journal.pntd. 0003756 PMID: 26061527; PubMed Central PMCID: PMC4464526.
- Senyonjo L, Aboe A, Bailey R, Agyemang D, Marfo B, Wanye S, et al. Operational adaptations of the trachoma pre-validation surveillance strategy employed in Ghana: a qualitative assessment of successes and challenges. Infect Dis Poverty. 2019; 8(1):78. Epub 2019/08/29. https://doi.org/10.1186/ s40249-019-0585-x PMID: 31455431; PubMed Central PMCID: PMC6712645.
- World Health Organisation. Ghana eliminates trachoma, freeing millions from suffering and blindness. Brazzaville, Accra & Geneva: World Health Organization, 2018.

- Mari L, Gatto M, Ciddio M, Dia ED, Sokolow SH, De Leo GA, et al. Big-data-driven modeling unveils country-wide drivers of endemic schistosomiasis. Sci Rep. 2017; 7(1):489. https://doi.org/10.1038/ s41598-017-00493-1 PMID: 28352101; PubMed Central PMCID: PMC5428445.
- Irvine MA, Kazura JW, Hollingsworth TD, Reimer LJ. Understanding heterogeneities in mosquito-bite exposure and infection distributions for the elimination of lymphatic filariasis. Proc Biol Sci. 2018; 285 (1871). https://doi.org/10.1098/rspb.2017.2253 PMID: 29386362; PubMed Central PMCID: PMC5805933.
- Truscott JE, Turner HC, Farrell SH, Anderson RM. Soil-Transmitted Helminths: Mathematical Models of Transmission, the Impact of Mass Drug Administration and Transmission Elimination Criteria. Advances in parasitology. 2016; 94:133–98. https://doi.org/10.1016/bs.apar.2016.08.002 PMID: 27756454.
- Greenhouse B, Daily J, Guinovart C, Goncalves B, Beeson J, Bell D, et al. Priority use cases for antibody-detecting assays of recent malaria exposure as tools to achieve and sustain malaria elimination. Gates Open Res. 2019; 3:131. https://doi.org/10.12688/gatesopenres.12897.1 PMID: <u>31172051</u>; PubMed Central PMCID: PMC6545519.
- Drakeley CJ, Corran PH, Coleman PG, Tongren JE, McDonald SL, Carneiro I, et al. Estimating medium- and long-term trends in malaria transmission by using serological markers of malaria exposure. Proceedings of the National Academy of Sciences of the United States of America. 2005; 102 (14):5108–13. <u>https://doi.org/10.1073/pnas.0408725102</u> PMID: <u>15792998</u>; PubMed Central PMCID: PMC555970.
- Corran P, Coleman P, Riley E, Drakeley C. Serology: a robust indicator of malaria transmission intensity? Trends in parasitology. 2007; 23(12):575–82. https://doi.org/10.1016/j.pt.2007.08.023 PMID: 17988945.
- Pinsent A, Solomon AW, Bailey RL, Bid R, Cama A, Dean D, et al. The utility of serology for elimination surveillance of trachoma. Nature communications. 2018; 9(1):5444. https://doi.org/10.1038/s41467-018-07852-0 PMID: 30575720; PubMed Central PMCID: PMC6303365.
- Arnold BF, van der Laan MJ, Hubbard AE, Steel C, Kubofcik J, Hamlin KL, et al. Measuring changes in transmission of neglected tropical diseases, malaria, and enteric pathogens from quantitative antibody levels. PLoS neglected tropical diseases. 2017; 11(5):e0005616. https://doi.org/10.1371/journal.pntd. 0005616 PMID: 28542223; PubMed Central PMCID: PMC5453600.
- Arnold BF, Martin DL, Juma J, Mkocha H, Ochieng JB, Cooley GM, et al. Enteropathogen antibody dynamics and force of infection among children in low-resource settings. Elife. 2019;8. <u>https://doi.org/</u> 10.7554/eLife.45594 PMID: 31424386; PubMed Central PMCID: PMC6746552.
- Martin DL, Bid R, Sandi F, Goodhew EB, Massae PA, Lasway A, et al. Serology for trachoma surveillance after cessation of mass drug administration. PLoS neglected tropical diseases. 2015; 9(2): e0003555. https://doi.org/10.1371/journal.pntd.0003555 PMID: 25714363; PubMed Central PMCID: PMC4340913.
- Mladonicky JM, King JD, Liang JL, Chambers E, Pa'au M, Schmaedick MA, et al. Assessing transmission of lymphatic filariasis using parasitologic, serologic, and entomologic tools after mass drug administration in American Samoa. The American journal of tropical medicine and hygiene. 2009; 80(5):769–73. PMID: 19407122.
- Lammie PJ, Moss DM, Brook Goodhew E, Hamlin K, Krolewiecki A, West SK, et al. Development of a new platform for neglected tropical disease surveillance. International journal for parasitology. 2012; 42 (9):797–800. https://doi.org/10.1016/j.ijpara.2012.07.002 PMID: 22846784.
- Arnold BF, Scobie HM, Priest JW, Lammie PJ. Integrated Serologic Surveillance of Population Immunity and Disease Transmission. Emerging infectious diseases. 2018; 24(7):1188–94. <u>https://doi.org/10.3201/eid2407.171928</u> PMID: 29912680; PubMed Central PMCID: PMC6038749.
- 20. Solomon AW, Engels D, Bailey RL, Blake IM, Brooker S, Chen JX, et al. A diagnostics platform for the integrated mapping, monitoring, and surveillance of neglected tropical diseases: rationale and target product profiles. PLoS neglected tropical diseases. 2012; 6(7):e1746. Epub 2012/08/04. https://doi.org/10.1371/journal.pntd.0001746 PMID: 22860146; PubMed Central PMCID: PMC3409112.
- Ashton RA, Kefyalew T, Rand A, Sime H, Assefa A, Mekasha A, et al. Geostatistical modeling of malaria endemicity using serological indicators of exposure collected through school surveys. The American journal of tropical medicine and hygiene. 2015; 93(1):168–77. https://doi.org/10.4269/ajtmh.14-0620 PMID: 25962770; PubMed Central PMCID: PMC4497890.
- Fronterre C, Amoah B, Giorgi E, Stanton MC, Diggle PJ. Design and Analysis of Elimination Surveys for Neglected Tropical Diseases. The Journal of infectious diseases. 2020. <u>https://doi.org/10.1093/infdis/jiz554</u> PMID: 31930383.

- Metcalf CJ, Farrar J, Cutts FT, Basta NE, Graham AL, Lessler J, et al. Use of serological surveys to generate key insights into the changing global landscape of infectious disease. Lancet. 2016; 388 (10045):728–30. https://doi.org/10.1016/S0140-6736(16)30164-7 PMID: 27059886.
- Arnold BF, Kanyi H, Njenga SM, Rawago FO, Priest JW, Secor WE, et al. Fine-scale heterogeneity in Schistosoma mansoni force of infection measured through antibody response. Proceedings of the National Academy of Sciences. 2020:202008951. https://doi.org/10.1073/pnas.2008951117 PMID: 32868437
- Helb DA, Tetteh KK, Felgner PL, Skinner J, Hubbard A, Arinaitwe E, et al. Novel serologic biomarkers provide accurate estimates of recent Plasmodium falciparum exposure for individuals and communities. Proceedings of the National Academy of Sciences of the United States of America. 2015; 112(32): E4438–47. https://doi.org/10.1073/pnas.1501705112 PMID: 26216993; PubMed Central PMCID: PMC4538641.
- Azman AS, Lessler J, Luquero FJ, Bhuiyan TR, Khan AI, Chowdhury F, et al. Estimating cholera incidence with cross-sectional serology. Sci Transl Med. 2019; 11(480). https://doi.org/10.1126/ scitranslmed.aau6242 PMID: 30787170; PubMed Central PMCID: PMC6430585.
- Debrah O, Mensah EO, Senyonjo L, de Souza DK, Hervie TE, Agyemang D, et al. Elimination of trachoma as a public health problem in Ghana: Providing evidence through a pre-validation survey. PLoS neglected tropical diseases. 2017; 11(12):e0006099. https://doi.org/10.1371/journal.pntd.0006099 PMID: 29232708; PubMed Central PMCID: PMC5746280.
- Senyonjo LG, Debrah O, Martin DL, Asante-Poku A, Migchelsen SJ, Gwyn S, et al. Serological and PCR-based markers of ocular Chlamydia trachomatis transmission in northern Ghana after elimination of trachoma as a public health problem. PLoS neglected tropical diseases. 2018; 12(12):e0007027. https://doi.org/10.1371/journal.pntd.0007027 PMID: 30550537; PubMed Central PMCID: PMC6310292.
- Solomon AW, Kurylo E. The global trachoma mapping project. Community Eye Health. 2014; 27 (85):18. PMID: 24966461; PubMed Central PMCID: PMC4069783.
- Sepulveda N, Stresman G, White MT, Drakeley CJ. Current Mathematical Models for Analyzing Anti-Malarial Antibody Data with an Eye to Malaria Elimination and Eradication. J Immunol Res. 2015; 2015:738030. <u>https://doi.org/10.1155/2015/738030</u> PMID: 26770994; PubMed Central PMCID: PMC4684866.
- Benagalia T, Chauveau D, Hunter DR, Young DS. mixtools: an R Package for Analyzing Mixture Models. Journal of Statistical Software. 2010; 32(6).
- 32. Surendra H, Supargiyono, Ahmad RA, Kusumasari RA, Rahayujati TB, Damayanti SY, et al. Using health facility-based serological surveillance to predict receptive areas at risk of malaria outbreaks in elimination areas. BMC Med. 2020; 18(1):9. https://doi.org/10.1186/s12916-019-1482-7 PMID: 31987052; PubMed Central PMCID: PMC6986103.
- 33. Ghinai R, El-Duah P, Chi KH, Pillay A, Solomon AW, Bailey RL, et al. A cross-sectional study of 'yaws' in districts of Ghana which have previously undertaken azithromycin mass drug administration for trachoma control. PLoS neglected tropical diseases. 2015; 9(1):e0003496. https://doi.org/10.1371/journal. pntd.0003496 PMID: 25632942; PubMed Central PMCID: PMC4310597.
- Lloyd CT, Sorichetta A, Tatem AJ. High resolution global gridded data for use in population studies. Sci Data. 2017; 4:170001. <u>https://doi.org/10.1038/sdata.2017.1</u> PMID: <u>28140386</u>; PubMed Central PMCID: PMC5283062.
- Weiss DJ, Nelson A, Gibson HS, Temperley W, Peedell S, Lieber A, et al. A global map of travel time to cities to assess inequalities in accessibility in 2015. Nature. 2018; 553(7688):333–6. https://doi.org/10. 1038/nature25181 PMID: 29320477.
- Hengl T, Heuvelink GBM, Kempen B, Leenaars JGB, Walsh MG, Shepherd KD, et al. Mapping Soil Properties of Africa at 250m Resolution: Random Forests Significantly Improve Current Predictions. PLoS One. 2015; 10(6):e0125814. https://doi.org/10.1371/journal.pone.0125814 PMID: 26110833
- Hansen MC, Potapov PV, Moore R, Hancher M, Turubanova SA, Tyukavina A, et al. High-resolution global maps of 21st-century forest cover change. Science. 2013; 342(6160):850–3. <u>https://doi.org/10.1126/science.1244693</u> PMID: 24233722.
- Fick SE, Hijmans RJ. Worldclim 2: New 1-km spatial resolution climate surfaces for global land areas. International Journal of Climatology. 2017.
- NASA/METI/AIST/Japan Spacesystems. Shuttle Radar Topography Mission (SRTM). NASA EOSDIS Land Processes DAAC2020.
- 40. OpenStreetMap Contributors. Open Street Map. OpenStreetMap Foundation; 2019.
- Landsat 8 Operational Land Imager [Internet]. NASA EOSDIS Land Processes DAAC, USGS Earth Resources Observation and Science (EROS) Center. 2014. Available from: http://landsat.usgs.gov// index.php.

- 42. UNEP-WCMC, IUCN. Protected Planet: The World Database on Protected Areas (WDPA). In: Nature UNEPWCMCalUfCo, editor. Cambridge, UK2019.
- 43. Simpson DP, Illian JB, Lindren F, Sorbye SH, Rue H. Penalising model component complexity: a principled, practical approach to constructing priors. Statistical Science. 2017.
- Rue H, Martino S, Chopin N. Approximate Bayesian inference for latent Gaussian models by using integrated nested Laplace approximations. Journal of the Royal Statistical Society. 2009;71(Part 2):319– 92.
- 45. Giorgi E, Diggle PJ, Snow RW, Noor AM. Geostatistical methods for disease mapping and visualisation using data from spatio-temporally referenced prevalence surveys. International Statistical Review. 2018; 86(3):571–97. https://doi.org/10.1111/insr.12268 PMID: 33184527
- 46. Stresman GH, Giorgi E, Baidjoe A, Knight P, Odongo W, Owaga C, et al. Impact of metric and sample size on determining malaria hotspot boundaries. Sci Rep. 2017; 7:45849. <u>https://doi.org/10.1038/srep45849 PMID: 28401903</u>; PubMed Central PMCID: PMC5388846.
- Salje H, Cummings DAT, Rodriguez-Barraquer I, Katzelnick LC, Lessler J, Klungthong C, et al. Reconstruction of antibody dynamics and infection histories to evaluate dengue risk. Nature. 2018; 557 (7707):719–23. https://doi.org/10.1038/s41586-018-0157-4 PMID: 29795354; PubMed Central PMCID: PMC6064976.
- Longley RJ, White MT, Takashima E, Brewster J, Morita M, Harbers M, et al. Development and validation of serological markers for detecting recent Plasmodium vivax infection. Nat Med. 2020; 26(5):741– 9. https://doi.org/10.1038/s41591-020-0841-4 PMID: 32405064.
- 49. Araujo Navas AL, Hamm NA, Soares Magalhaes RJ, Stein A. Mapping Soil Transmitted Helminths and Schistosomiasis under Uncertainty: A Systematic Review and Critical Appraisal of Evidence. PLoS neglected tropical diseases. 2016; 10(12):e0005208. https://doi.org/10.1371/journal.pntd.0005208 PMID: 28005901; PubMed Central PMCID: PMC5179027.
- 50. Njenga SM, Kanyi HM, Arnold BF, Matendechero SH, Onsongo JK, Won KY, et al. Integrated Cross-Sectional Multiplex Serosurveillance of IgG Antibody Responses to Parasitic Diseases and Vaccines in Coastal Kenya. The American journal of tropical medicine and hygiene. 2020; 102(1):164–76. https:// doi.org/10.4269/ajtmh.19-0365 PMID: 31769388; PubMed Central PMCID: PMC6947807.
- Kucharski AJ, Kama M, Watson CH, Aubry M, Funk S, Henderson AD, et al. Using paired serology and surveillance data to quantify dengue transmission and control during a large outbreak in Fiji. Elife. 2018;7. https://doi.org/10.7554/eLife.34848 PMID: 30103854; PubMed Central PMCID: PMC6092126.